Author's response to reviews

Title: Poor mental health and sexual risk behaviours in Uganda: A cross-sectional population-based study

Authors:

Patric Lundberg (patric.lundberg@ki.se)
Godfrey Rukundo (rukundogzari@yahoo.co.uk)
Schola Ashaba (scholaashaba@yahoo.com)
Anna Thorson (anna.thorson@ki.se)
Peter Allebeck (peter.allebeck@ki.se)
Per-Olof Östergren (Per-Olof.Ostergren@med.lu.se)
Elizabeth Cantor-Graae (elizabeth.cantor-graae@med.lu.se)

Version: 7 Date: 18 February 2011

Author's response to reviews: see over
Dear Sir,

During the proofing process we realised that an important study on mental health and sexual risk behaviours in a middle income country has been published after we submitted our manuscript in November:


In order to improve our manuscript we would like to ask for your permission to update the manuscript so this reference can be included, if possible.

We are therefore now resubmitting a new version of the manuscript where this reference has been included. We have changed the text accordingly in the introduction (See page 4, last paragraph, ‘Nevertheless…’).

We have also added this new reference to the discussion section (See page 18, the second last paragraph, the sentence starting with ‘Based on our findings…’), although no text has been changed in the discussion section.

In addition we have made one correction: The last reference in the introduction was incorrect and has been changed (See page 5, reference 27). No text has been changed.

Yours sincerely,

Patric Lundberg